Author's response to reviews

Title: Diagnosis of tuberculosis in an Indian population by an indirect ELISA protocol based on detection of Antigen 85 complex: a prospective cohort study

Authors:

Rajpal S. Kashyap (raj_ciims@rediffmail.com)
Anju N Rajan (anju_rajann@rediffmail.com)
Sonali S Ramteke (sonaliramteke@yahoo.co.in)
Vijay S. Agrawal (ciims_ngp@sancharnet.in)
Sanjivani S Kelkar (ciims_ngp@sancharnet.in)
Hemant J. Purohit (hemantdrd@hotmail.com)
Girdhar M. Taori (taorigm_ciims@yahoo.co.in)
Hatim F. Daginawala (hfd_ciims@rediffmail.com)

Version: 3 Date: 18 January 2007

Author's response to reviews:

Correction made:

1. Figure 2 is included in the result section of the manuscript.